Author’s response to reviews

Title: Survival status and factors associated with treatment outcome of severely malnourished children admitted to Ayder Referral Hospital: A cross-sectional study

Authors:

Haftay Mezgebe (haftay.berhane@mu.edu.et)

Tesfay Atey (tesmar11@gmail.com)

Mengistu Girma (mengistugirma427@yahoo.com)

Version: 4 Date: 09 Jun 2017

Author’s response to reviews:

On behalf of the authors, I am very glad to see our manuscript published in your journal. We have added the statement `The importance of the study, data collection process, confidentiality and the ethical issue was briefly described for parents of the study participants before data collection, and verbal consent was obtained from parents of each child patient` to satisfy the comment provided.

We have also added a "Figure legends" section after the reference